Author's response to reviews

Title: The SHED-IT community trial study protocol: a randomised controlled trial of weight loss programs for overweight and obese men

Authors:

Philip J Morgan (Philip.Morgan@newcastle.edu.au)
Clare E Collins (Clare.Collins@newcastle.edu.au)
Ronald C Plotnikoff (Ron.Plotnikoff@newcastle.edu.au)
Patrick McElduff (Patrick.McElduff@newcastle.edu.au)
Tracy Burrows (Tracy.Burrows@newcastle.edu.au)
Janet M Warren (janet.warren@danone.com)
Myles D Young (Myles.Young@newcastle.edu.au)
Nina Berry (Nina.Berry@newcastle.edu.au)
Kristen L Saunders (Kristen.Saunders@newcastle.edu.au)
Elroy J Aguiar (Elroy.Aguiar@newcastle.edu.au)
Robin Callister (Robin.Callister@newcastle.edu.au)

Version: 5 Date: 15 November 2010

Author's response to reviews:

After a final proofing, all typographical errors have been amended and a few sentences have been reworded to improve readability.